Author's response to reviews

Title: Linkage disequilibrium mapping of a breast cancer susceptibility locus in the RAI gene region

Authors:

Bjorn A Nexo (nexo@humgen.au.dk)
Ulla Vogel (ubv@nrcwe.dk)
Anja Olsen (anja@cancer.dk)
Mette Nyegaard (nyegaard@humgen.au.dk)
Zuzanna Bukowy (zb@ibb.waw.pl)
Eszter Rockenbauer (rockenbauer@hotmail.com)
Xiuqing Zhang (zhangxq@genomics.org.cn)
Cemile Koca (cemile_md@yahoo.com)
Mette Mains (mette@mains.dk)
Bettina Hansen (bhansen@humgen.au.dk)
Anne Hedemand (anneh@humgen.au.dk)
Anette Kjeldgaard (anettekjeldgaard@hotmail.com)
Magdalena J Laska (laska@humgen.au.dk)
Ole Raaschou-Nielsen (ole@cancer.dk)
Soren Cold (soeren.cold@ouh.fyns-amt.dk)
Kim Overvad (ko@dce.au.dk)
Anne Tjonneland (annet@cancer.dk)
Lars Bolund (bolund@humgen.au.dk)
Anders D Borglum (anders@humgen.au.dk)

Version: 2 Date: 11 July 2007

Author's response to reviews: see over
Uploaded, please find a revised version of the manuscript “Nexø et al: Linkage disequilibrium mapping of a breast cancer susceptibility locus in the RAI gene region”, which we hereby submit to your journal for publication.

The only change relative to the manuscript previously forwarded to BMC Genomics is the inclusion of a paragraph in the Materials and Methods section specifying our compliance with the Helsinki declaration, our approval by a Science Ethics Committee, and the fact that we have obtained informed consent from all participants.

The experimental data and the manuscript is our work, and they are not in consideration for publication elsewhere.

Yours sincerely

Bjørn Nexø